International Journal of Linguistics, Literature, and Culture



REVIEW HISTORY

Paper: "The Pragmatics of Email Request between Iraqi Students and English

Native Speakers"

Corresponding Author: Raad Hameed Kambash

Email: ayad.h.english@nuc.edu.iq

Doi: 10.19044/llc.v7no1a1

Peer review:

Reviewer 1: Saeid Rahimipoor Farhangian University, Iran

Reviewer 2: Balanescu Olga Carmen University of Bucharest, Romania

Published: 31.03.2020

